

## Format for ANSWERING REVIEWERS

October15, 2014

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 13104-review.doc).

Title: Technical skills and training of upper gastrointestinal endoscopy for new beginners

Author: Seung-Hwa Lee, Young-Kyu Park, Sung-Min Cho, Joon-Koo Kang, Duck-Joo Lee

Name of Journal: World Journal of Gastroenterology

## ESPS Manuscript NO: 13104

The manuscript has been improved according to the suggestions of reviewers:

1.Format has been updated

2. Revision has been made according to the suggestions of the reviewer

 $\rightarrow$  We deeply appreciate reviewers for their elaborate comments on our manuscript. We tried our best to mend and ameliorate our manuscript sincerely following reviewer's precious advices.

(1) Review for the manuscript No: 13104 Technical Skills and Training of Upper Gastrointestinal Endoscopy for Beginners General comments: This is a very detailed and nice overview about Upper Endoscopy with a lot of endoscopic pictures. The manuscript is prepared on 123 pages, therefore I think in a such comprehensive form, it is not suitable for publication in WJG.

 $\rightarrow$  We sincerely appreciate your careful comment. According your advice, we revised main text of the manuscript to make it more clear, logical, and concise. Also, we transfer all figures into appendix. We believe that all figures in the manuscript will be helpful to new beginners.

(2) 1. Manuscript is too long. Should be shortened, 2. Some grammar and spelling mistakes

 $\rightarrow$  We sincerely appreciate your careful comment. According your advice, we revised main text of the manuscript to make it more clear, logical, and concise. We also regret that some of the sentences and phrases in our manuscript were rigid and incorrect but rigid. Hence, we revised completely our manuscript with the guidance of a native English speaker. We sincerely tried our best to edit and ameliorate our manuscript.

3.References and typesetting were corrected

Thank you again for publishing our manuscript in the World Journal of Gastroenterology.

Sincerely yours,

Young-Kyu Park, Professor

Center for Health Promotion Bundang Jesaeng Hospital Kwandong University School of Medicine Seongnam,Gyeonggi-do463-774,South Korea Email: drlovedd@hanmail.net